Author’s response to reviews

Title: The Use of Hyperbaric Oxygen Therapy in the Treatment of Central Retinal Artery Occlusion in Sickle Cell Disease.

Authors:

Handan Canan (handanakkaya@yahoo.com)
Burak Ulas (drburakulas@gmail.com)
Rana Altan-Yaycioglu (raltanya@yahoo.com)

Version: 2 Date: 8 May 2014

Author’s response to reviews:

Thank you for your interest on our paper. We have made the necessary changes on the paper.
Sincerely yours,